# A facilitated home-based cardiac rehabilitation intervention for people with heart failure and their caregivers: a research programme including the REACH-HF RCT

Hasnain M Dalal,<sup>1,2\*</sup> Rod S Taylor,<sup>1,3</sup> Jennifer Wingham,<sup>1</sup> Colin J Greaves,<sup>4</sup> Kate Jolly,<sup>5</sup> Chim C Lang,<sup>6</sup> Russell C Davis,<sup>7</sup> Karen M Smith,<sup>8</sup> Patrick J Doherty,<sup>9</sup> Jackie Miles,<sup>10</sup> Robin van Lingen,<sup>11</sup> Fiona C Warren,<sup>1</sup> Susannah Sadler,<sup>1</sup> Charles Abraham,<sup>1,12</sup> Nicky Britten,<sup>1</sup> Julia Frost,<sup>1</sup> Melvyn Hillsdon,<sup>13</sup> Sally Singh,<sup>14</sup> Christopher Hayward,<sup>15</sup> Victoria Eyre<sup>16</sup> and Kevin Paul<sup>1,2</sup>

- <sup>1</sup>Institute of Health Research, University of Exeter Medical School, Exeter, UK
- <sup>2</sup>Research and Development, Royal Cornwall Hospitals NHS Trust, Truro, UK
- <sup>3</sup>Professor of Population, Institute of Health and Wellbeing, University of Glasgow, Glasgow, UK
- <sup>4</sup>School of Sport, Exercise and Rehabilitation Sciences, University of Birmingham, Birmingham, UK
- <sup>5</sup>Institute of Applied Health Research, University of Birmingham, Birmingham, UK
- <sup>6</sup>School of Medicine, University of Dundee, Ninewells Hospital and Medical School, Dundee, UK
- <sup>7</sup>Cardiology Department, Sandwell and West Birmingham Hospitals NHS Trust, Birmingham, UK
- <sup>8</sup>Cardiac Rehabilitation Office, Ninewells Hospital, Dundee, UK
- <sup>9</sup>Department of Health Sciences, University of York, York, UK
- <sup>10</sup>Research and Development, Aneurin Bevan University Health Board, St Woolos Hospital, Newport, UK
- <sup>11</sup>Duchy Hospital, Truro, UK
- <sup>12</sup>School of Psychological Sciences, University of Melbourne, Victoria, VIC, Australia <sup>13</sup>Sport and Health Sciences, University of Exeter, Exeter, UK
- <sup>14</sup>Centre for Exercise and Rehabilitation Science, University Hospitals of Leicester NHS Trust, Glenfield Hospital, Leicester, UK
- <sup>15</sup>Peninsula Clinical Trials Unit, University of Plymouth, Plymouth, UK <sup>16</sup>Re:Cognition Health, London, UK

\*Corresponding author h.dalal@nhs.net

**Declared competing interests of authors:** Hasnain M Dalal is an ordinary council member of the British Association of Cardiovascular Prevention and Rehabilitation. Rod S Taylor is a member of the Health Technology Assessment (HTA) National Stakeholder Advisory Group. Kate Jolly reports grants from the National Institute for Health Research (NIHR) Public Health Research, NIHR Research for Patient Benefit, NIHR Global Health Research Group, NIHR HTA and NIHR Collaborations for Leadership in Applied Health Research and Care programme outside the submitted work. Russell C Davis reports personal fees from Novartis International AG (Basel, Switzerland), Servier Laboratories (Suresnes, France) and Bristol Myers Squibb (New York, NY, USA) outside the submitted work. Charles Abraham, Rod S Taylor, Colin J Greaves and Nicky Britten are also supported by the NIHR Collaborations for Leadership in Applied Health Research and Care (CLAHRC) South West Peninsula at the Royal Devon and Exeter NHS Foundation Trust; Kate Jolly by CLAHRC West Midlands; and Sally Singh by CLAHRC East Midlands. For Charles Abraham and Nicky Britten, this work was partially funded by the UK NIHR PenCLAHRC. Rod S Taylor, Sally Singh and Hasnain M Dalal report receiving research funding for trials of ExCR during the course of this updated review. All authors report grants from NIHR during the course of the trial.

Published February 2021 DOI: 10.3310/pgfar09010

## **Plain English summary**

A facilitated home-based cardiac rehabilitation intervention Programme Grants for Applied Research 2021; Vol. 9: No. 1 DOI: 10.3310/pgfar09010

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

eart failure can be classified by the percentage of blood pushed out of the left ventricle when the heart beats – the 'ejection fraction'. Half of all people with heart failure have heart failure with reduced ejection fraction (ejection fraction of < 45%) and the rest have heart failure with preserved ejection fraction (ejection fraction of  $\geq$  45%).

Rehabilitation for people with heart failure improves their quality of life and reduces hospital admissions. Most rehabilitation has taken place in groups in hospital settings, but poor levels of participation mean that alternative ways of accessing rehabilitation, such as home-based programmes, are recommended.

The Rehabilitation Enablement in Chronic Heart Failure (REACH-HF) intervention comprises three printed components: (1) an 'HF Manual' with information for patients and an exercise programme, (2) a 'Family and Friends Resource' for caregivers and (3) a 'Progress Tracker' booklet for patients to chart their progress. The programme is delivered by trained nurses or physiotherapists.

We carried out two studies of the REACH-HF intervention. In our main trial, we allocated at random 216 people with heart failure with reduced ejection fraction from four UK centres to receive the REACH-HF intervention or usual medical care alone. The results of this study suggest that those who received the REACH-HF intervention had a better quality of life. Economic modelling demonstrated that the intervention is well within the willingness-to-pay limits recommended by the National Institute for Health and Care Excellence. However, the economics analysis from the full trial did not show significant differences between the two groups in hospital admissions. The cost of the intervention was £418 per patient.

We also undertook a pilot trial of the same design in a single centre in 50 patients with heart failure with preserved ejection fraction. The findings of this pilot study showed that the intervention and study design were well accepted and these results support a future full trial of REACH-HF in patients with heart failure with preserved ejection fraction.

### **Programme Grants for Applied Research**

ISSN 2050-4322 (Print)

ISSN 2050-4330 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full PGfAR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/pgfar. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Programme Grants for Applied Research journal

Reports are published in *Programme Grants for Applied Research* (PGfAR) if (1) they have resulted from work for the PGfAR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **Programme Grants for Applied Research programme**

The Programme Grants for Applied Research (PGfAR) programme, part of the National Institute for Health Research (NIHR), was established in 2006 to fund collaborative, multidisciplinary programmes of applied research to solve health and social care challenges. Findings are expected to provide evidence that lead to clear and identifiable patient benefits, in the relatively near future.

PGfAR is researcher led and does not specify topics for research; however, the research must be in an area of priority or need for the NHS and the social care sector of the Department of Health and Social Care, with particular emphasis on health and social care areas that cause significant burden, where other research funders may not be focused, or where insufficient funding is available.

The programme is managed by the NIHR Central Commissioning Facility (CCF) with strategic input from the Programme Director. For more information about the PGfAR programme please visit the website: https://www.nihr.ac.uk/explore-nihr/funding-programmes/programme-grants-for-applied-research.htm

#### This report

The research reported in this issue of the journal was funded by PGfAR as project number RP-PG-1210-12004. The contractual start date was in January 2013. The final report began editorial review in March 2019 and was accepted for publication in November 2020. As the funder, the PGfAR programme agreed the research questions and study designs in advance with the investigators. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. This report has been published following a shortened production process and, therefore, did not undergo the usual number of proof stages and opportunities for correction. The PGfAR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, CCF, NETSCC, PGfAR or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the PGfAR programme or the Department of Health and Social Care.

Copyright © 2021 Dalal *et al.* This work was produced by Dalal *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### NIHR Journals Library Editors

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk